

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Perez-Vincent, Santiago M.; Schargrodsky, Ernesto; García Mejía, Mauricio

Working Paper

Crime under lockdown: The impact of COVID-19 on citizen security in the city of Buenos Aires

IDB Working Paper Series, No. IDB-WP-1245

Provided in Cooperation with: Inter-American Development Bank (IDB), Washington, DC

Suggested Citation: Perez-Vincent, Santiago M.; Schargrodsky, Ernesto; García Mejía, Mauricio (2021) : Crime under lockdown: The impact of COVID-19 on citizen security in the city of Buenos Aires, IDB Working Paper Series, No. IDB-WP-1245, Inter-American Development Bank (IDB), Washington, DC, https://doi.org/10.18235/0003431

This Version is available at: https://hdl.handle.net/10419/237508

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

ND https://creativecommons.org/licenses/by-nc-nd/3.0/igo/legalcode







IDB WORKING PAPER SERIES N° IDB-WP-1245

Crime under Lockdown

The Impact of COVID-19 on Citizen Security in the City of Buenos Aires

Santiago M. Perez-Vincent Ernesto Schargrodsky Mauricio García Mejía

Inter-American Development Bank Institutions for Development Sector

August 2021



Crime under Lockdown

The Impact of COVID-19 on Citizen Security in the City of Buenos Aires

Santiago M. Perez-Vincent Ernesto Schargrodsky Mauricio García Mejía

August 2021

Cataloging-in-Publication data provided by the Inter-American Development Bank Felipe Herrera Library

Perez-Vincent, Santiago M.

Crime under lockdown: the impact of COVID-19 on citizen security in the city of Buenos Aires / Santiago M. Perez-Vincent, Ernesto Schargrodsky, Mauricio García Mejía. p. cm. — (IDB Working Paper Series ; 1245)

Includes bibliographic references.

1. Criminal statistics-Argentina. 2. Crime prevention-Argentina. 3. Public safety-Argentina.

 Coronavirus infections-Social aspects-Argentina.
Quarantine-Social aspects-Argentina.
Schargrodsky, Ernesto. II. García Mejía, Mauricio. III. Inter-American Development Bank. Innovation in Citizen Services Division. IV. Title.
V. Series.
IDB-WP-1245

http://www.iadb.org

Copyright © 2021 Inter-American Development Bank. This work is licensed under a Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives (CC-IGO BY-NC-ND 3.0 IGO) license (<u>http://creativecommons.org/licenses/by-nc-nd/3.0/igo/</u> <u>legalcode</u>) and may be reproduced with attribution to the IDB and for any non-commercial purpose, as provided below. No derivative work is allowed.

Any dispute related to the use of the works of the IDB that cannot be settled amicably shall be submitted to arbitration pursuant to the UNCITRAL rules. The use of the IDB's name for any purpose other than for attribution and the use of IDB's logo shall be subject to a separate written license agreement between the IDB and the user and is not authorized as part of this CC-IGO license.

Following a peer review process, and with previous written consent by the Inter-American Development Bank (IDB), a revised version of this work may also be reproduced in any academic journal, including those indexed by the American Economic Association's EconLit, provided that the IDB is credited and that the author(s) receive no income from the publication. Therefore, the restriction to receive income from such publication shall only extend to the publication's author(s). With regard to such restriction, in case of any inconsistency between the Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives license and these statements, the latter shall prevail.

Note that the link provided above includes additional terms and conditions of the license.

The opinions expressed in this publication are those of the authors and do not necessarily reflect the views of the Inter-American Development Bank, its Board of Directors, or the countries they represent.



Crime under Lockdown: The Impact of COVID-19 on Citizen Security in the City of Buenos Aires

Santiago M. Perez-Vincent

IDВ

Ernesto Schargrodsky UTDT, CAF and CONICET Mauricio García Mejía *IDB*

Abstract*

This paper studies the impact of the COVID-19 pandemic and the subsequent lockdown on criminal activity in the City of Buenos Aires, Argentina. We find a large, significant, robust, and immediate decline in crime following quarantine restrictions. We observe the effect on property crime reported to official agencies, police arrests, and crime reported in victimization surveys, but not in homicides. The decrease in criminal activity was greater in business and transportation areas, but still large in commercial and residential areas (including informal settlements). After the sharp and immediate fall, crime recovered but, as of November 2020, it did not reach its initial levels. The arrest data additionally allow us to measure the distance from the detainee's address to the crime location. Crime became more local as mobility was restricted.

JEL Codes: J18, K42 Keywords: citizen security, crime, Argentina, COVID-19, lockdown

^{*} Santiago M. Perez-Vincent: santiagoper@iadb.org; Ernesto Schargrodsky: eschargr@utdt.edu; Mauricio García Mejía: mauriciogar@iadb.org. We thank Anibal Falivene, Carola Jersonsky, Mariana Cirese, Ana María Magri, María Paula Spagnoletti and Ezequiel Montana from the Ministry of Justice and Security of the City of Buenos Aires for guidance and data provision, Julián Regatky and Layla Such Nachtrieb for excellent research assistance, and the Inter-American Development Bank for financial support. The views expressed in this paper are those of the authors and do not necessarily reflect the views of the Inter-American Development Bank, its Board of Directors, or the countries they represent.

1. Introduction

The COVID-19 pandemic shook humanity to its core in 2020. In addition to its direct health effects, the pandemic forced worldwide authorities to implement extraordinary policies to curb the spread of the virus and prevent the healthcare system's collapse. These policies included quarantines, closure of non-essential services and businesses, interruption of face-to-face schooling, constraints on the use of public transportation, restrictions on gathering size, and comprehensive control and closure of international and domestic borders (Hale et al., 2020). Beyond these public policies, individuals adopted self-protection measures, such as maintaining social distancing and adjusting hygiene practices. The fear of contagion and the containment measures disrupted people's lives, changing almost all activity patterns and social dynamics.

Criminal activity was not an exception. The COVID-19 pandemic and the accompanying changes in social dynamics affected victims, criminals, police agencies, and criminal justice systems in multiple ways. These impacts altered opportunities and incentives for criminal activity through several simultaneous and heterogeneous channels.

First, the government-imposed quarantines and curfews, and the individuals' self-protection decisions had marked effects on movement patterns (see, for example, SLOCAT, 2020). People's schedules and family dynamics changed drastically: work was restricted—or completely suspended for non-essential activities—, schools were closed, public transportation was limited, and several social activities were banned, all of which significantly reduced the exposure of potential victims to out-of-home crime. Second, the pandemic and the lockdowns simultaneously affected supply and demand in most economic activities, leading to an unprecedented economic contraction (World Bank, 2020). The recession triggered unemployment and poverty, while the heterogeneity of the shocks increased inequality, likely affecting the economic incentives to get involved in criminal activities (Raphael and Winter-Ebmer, 2001; Fainzylber et al., 2002; Schargrodsky and Freira, 2021). Third, police dynamics changed by increasing its presence and visibility in the streets, while also altering its focus from crime prevention to the enforcement of mobility restrictions (Lum et al., 2020; Alvarado et al., 2020). This intensified police presence probably had a deterrence effect on criminals, but incapacitation effects might have weakened. Fourth, in several countries, inmates who were soon to serve their sentences or were at risk of COVID-19 complications were released or given house-arrest, to reduce prison overcrowding and contagion among inmates and prison officers (Londoño et al., 2020; ACLU Analytics, 2020). Moreover, detentions for some minor offenses were reduced or put on hold due to delays in the judicial systems, whose operating capacity was also affected by lockdowns. These changes in the criminal justice system's functioning might have reduced its incapacitation and deterrence effects, potentially contributing to more criminal activity. Finally, school closures and social programs' suspensions reduced access to local networks and support services among at-risk youth, potentially increasing their vulnerability to criminality.

In short, the pandemic affected victims, criminals, police forces, and the judiciary and penitentiary systems. The combination of several simultaneous and heterogeneous shocks calls for an

empirical answer to the question of the impact of the pandemic on crime and violence. This study discusses the pandemic's impact on crime in the City of Buenos Aires, the capital of Argentina.

Argentina's national government imposed a mandatory lockdown on March 20, 2020. In the City of Buenos Aires, the lockdown remained in place (with some modifications) until November 9, 2020. We use official historical statistics on crime and police arrests, and crime victimization survey data. We estimate the pandemic's overall impact on crime using a city-level linear regression model and controlling for secular trends and seasonal variations. Our identification of the pandemic's impact relies on the assumption that, had the pandemic not occurred, the different outcomes (such as the number of crimes) would have shown a seasonal evolution similar to that of the previous years. We assess the robustness of our results by estimating an alternative event-study model and using different econometric specifications.

We observe a large, significant, and immediate decline in crime following the imposition of the lockdown. After this sharp fall, crime started to recover, but it had not reached its pre-lockdown levels by the end of the strict lockdown in November 2020. On average, the lockdown led to a 52.5 percent fall in reported property crimes, and a 59.3 percent fall in police arrests. Moreover, crime victimization measured from a monthly victimization survey fell 20.7 percent. Instead, we find no significant change in the number of homicides. In addition, we observe a large and significant increase (84.6 percent) in the number of detentions due to "resistance to authorities." The decrease in criminal activity was greater in business and transportation areas but still large in commercial and residential areas. These latter areas include informal settlements (*barrios populares*), which show an additional fall in theft, burglary, and larceny when a test-and-trace program (the *Detectar* Program) was implemented, suggesting that increased government presence in these areas could have had positive externalities on crime.

We contribute to the novel literature on the impact of the COVID-19 pandemic on crime in different parts of the world. Several studies report an overall decline in crime during the first weeks of isolation in Los Angeles, United States. (Campedelli et al., 2020), San Francisco and Oakland, United States (Shayegh and Malpede, 2020); Vancouver, Canada (Hodgkinson and Andresen, 2020); Sweden (Gerell et al., 2020), Queensland, Australia (Andresen and Hodgkinson, 2020), and Mexico City, Mexico (Balmori de la Miyar et al., 2020).

The changes were not homogenous across different types of crime. Campedelli et al. (2020) found that in Los Angeles, non-violent property crimes fell to a greater extent. Halford et al. (2020), using data from a police district in the United Kingdom found that shoplifting and theft were the crimes that declined the most after the first week of lockdown. In Sweden, pickpocketing experienced the largest decline (Gerell et al., 2020). Rosenfeld and Lopez (2020) found that property and drug crime fell, while violent crime increased in a study examining 27 U.S. cities.

The evidence on the dynamics of serious assaults is mixed. Ashby (2020) and Campedelli et al. (2020) found no significant changes in reported assaults in several U.S. cities. However, there was a significant decline at least during some weeks in Sweden (Gerell et al., 2020), and Mexico

City (Balmori de la Miyar et al., 2020). Halford et al. (2020) also noticed a decrease in the first week of confinement in a police district in the United Kingdom.

The evidence regarding car thefts is also mixed. Los Angeles, San Francisco, and Denver saw an increase in car thefts during the pandemic's first wave, whereas there was a decrease in Atlanta, Detroit and Dallas, and no variation in Indianapolis (Mohler, 2020; Rosenfeld and Lopez, 2020). Hodgkinson et al. (2020) found no change in Vancouver's trend (when rates usually increase due to seasonal patterns), nor did Halford et al. (2020) for a police force area in the United Kingdom. For Queensland (Australia), Andresen and Hodgkinson (2020) found significant declines in most policing districts after the lockdown imposition, whereas Payne and Morgan (2020) found no major changes. In Latin America, there is evidence of a sharp fall in car thefts in urban areas of Colombia (Alvarado et al., 2020), and Mexico City (Balmori de la Miyar et al., 2020).

Most studies find a decline (albeit moderate) in reported burglaries after the start of the pandemic (Andresen and Hodgkinson, 2020; Ashby, 2020; Balmori de la Miyar et al., 2020; Halford et al., 2020; Mohler et al., 2020). The only exceptions are two studies covering Sweden and Queensland (Australia), where the number of reported burglaries remained unchanged (Gerell et al., 2020; Payne and Morgan, 2020). Residential burglary declined by 20 percent from March to June 2020 in several U.S. cities, but commercial burglaries spiked at the end of May in association with mass protests against police violence (Rosenfeld and Lopez, 2020). In Detroit, Felson et al. (2020) find a decline in burglaries in residential areas and an increase in mixed land use areas.

Homicides—arguably the most violent and costly crime—showed heterogeneous dynamics across periods and places. A study for 64 cities in the United States revealed an overall decline in monthly homicides rates during April and May 2020. However, rates rose in 25 of the 64 cities examined (Abt et al., 2020). Rosenfeld and Lopez (2020) reported an increase across 11 U.S. cities in June 2020, led by Chicago, Philadelphia, and Milwaukee. In Latin America, homicides fell in Peru (Calderon-Anyosa and Kaufman, 2020), El Salvador and Honduras (Semple and Ahmed, 2020) after the lockdown. The initial dynamics were similar in Colombia, where homicide rates fell sharply during the pandemic's first weeks. However, in June 2020, homicides had already returned to their pre-pandemic trend (Alvarado et al., 2020). In Mexico City, there was no significant change in homicide rates (Balmori de la Miyar et al., 2020).

Perhaps the most comprehensive study on the impact of the COVID-19 pandemic on crime is Nivette et al. (2021). This article considers 27 cities worldwide for six major crime categories (assault, theft, burglary, robbery, vehicle theft, and homicide). It finds that lockdowns were associated with a considerable, but heterogeneous drop in urban crime. Their overall results show declines in all types of crime, except for homicide. Our findings for the City of Buenos Aires of a large and significant decline in reported property crimes, police arrests, and crime victimization, with no relevant change in homicides, are in line with the results of the literature on the impact of the COVID-19 pandemic on crime.¹

Most studies of the COVID-19 pandemic's impact on crime rely only on official crime reports. However, the gap between actual and reported crime could be particularly relevant when assessing the crime impact of the pandemic. Mobility restrictions might have affected people's ability to report crimes. Moreover, the pandemic has also affected patrolling agencies, which have taken over new tasks (Lum et al., 2020). Lower police availability because of staff absences due to illness and safety measures to prevent police contagion may have affected reporting (Halford et al., 2020). A survey conducted across 13 police agencies in Latin America and the Caribbean reported that all agencies modified their proactive and preventive activities (Alvarado et al., 2020). Thus, the observed changes in reported crime rates during the pandemic might combine changes in actual crime levels and changes in reporting rates. Instead, our use of both official reports and a victimization survey allows us to address this potential shortcoming. In addition to extending the study of the crime impact of the COVID-19 pandemic to Buenos Aires, a large Latin American capital city, we contribute to this recent literature by showing that the observed fall in reported crime is not just an artifact of diminished reporting, but it is also observed in survey victimization data.

Moreover, our dataset's richness, which includes information on detainees' area of residence and crime location, also allows us to evaluate how the lockdown affected criminals' mobility patterns. providing insights into structural features of criminal activity. In addition to the reduction in overall crime, we find that crime became "more local" during the pandemic. The lockdown reduced the share of detainees from outside the City of Buenos Aires and decreased the distance from the detainee's area of residence to the crime location. This reduction in the share of detainees from outside the city, likely driven by increased controls in the city accesses, did not seem to lead to a relative augment in crime in the suburban areas. The lockdown's impact on reported victimization in the City's suburbs and the rest of the country was not statistically different from that inside the City of Buenos Aires. These results align with the hypothesis that focalized place-based interventions have the potential to reduce overall crime rates. These findings coincide with previous studies on local police deployments (see Di Tella and Schargrodsky, 2004, for Buenos Aires; and Draca et al., 2011, for London) and with the hot-spot literature (see Weisburd et al., 2012; Braga et al., 2014; Weisburd and Telep, 2016; Braga et al., 2019). Some recent studies, however, find that, because of the presence of displacement, hot-spot interventions or comparable natural experiments might not induce reductions in total crime (see Sherman et al., 2014, for Trinidad and Tobago; Collazos et al., 2020, for Medellin; Blattman et al., 2021, for Bogota; and Hodgkinson et al. 2020, for a Midwestern Canadian city). The shift in patrolling patterns and the strict mobility controls across jurisdictions imposed in Buenos Aires during the pandemic unfolds a proper scenario to assess the displacement of crime and to contribute to this open debate.

¹ Our study does not cover domestic and intrafamily violence. These crimes (or, at least, their reporting) displayed different dynamics across countries and cities. See Boxall et al. (2020) for Australia, Balmori de la Miyar et al. (2020) for Mexico City, and Perez-Vincent et al. (2020) for the City of Buenos Aires.

The rest of the paper is organized as follows. Section 2 describes the sequence of measures implemented by the Argentine authorities to address the pandemic. Section 3 details our data and empirical strategy. Section 4 presents the results for the different crime measures. Finally, section 5 summarizes the results and presents conclusions.

2. The Initial Impact of the COVID-19 Pandemic in Argentina and the City of Buenos Aires

The World Health Organization declared COVID-19 a pandemic on March 11, 2020 (WHO, 2020). Argentina's national government ordered its population to remain under a mandatory lockdown on March 20, 2020, closing international and domestic borders and only allowing essential businesses to remain open. The government enforced this initial phase of strict isolation until April 12. Then, each province and municipality periodically relaxed or strengthened mobility restrictions (alternating between strict lockdown phases and social distancing regimes) according to the number of positive COVID-19 cases and available intensive care unit (ICU) beds.

The City of Buenos Aires, the capital of Argentina, is an autonomous federal territory that does not belong to any of the country's 24 provinces. The city government maintained the strict initial restrictions until May 11 (GCBA, 2020a). After that day, different economic and recreational activities began to be progressively allowed. On June 8, the local government authorized recreational rides for children during weekends and nighttime outdoor physical activities (GCBA, 2020a). After an increase in positive COVID-19 cases, local authorities decided to reverse these relaxations and reinstated a strict lockdown between June 29 and July 17. After this day, the City of Buenos Aires began a staggered re-opening of different economic sectors (GCBA, 2020b). On November 9, the City of Buenos Aires transitioned from the mandatory lockdown phase to one of social distancing (GCBA, 2020c), in which mobility permits were no longer necessary. Schools began to re-open, and the government permitted more social gatherings (GCBA, 2020d). Figure 1 illustrates the timing of these phases.

Figure 1. COVID-19 Policies and Restrictions in the City of Buenos Aires, March– November 2020



Sources: WHO (2020), GCBA (2020a), GCBA (2020b), GCBA (2020c), GCBA (2020d).

Mobility fell very sharply throughout the city after the lockdown began and started to recover slowly. The number of public transportation users doubled in early May 2020 compared to the first week of isolation (La Nación, 2020a). Despite this recovery, mobility remained well below prelockdown levels for several months. National authorities introduced a unique mobility permit and a mobile phone application ("Cuidar") to extend mobility permits, monitor citizens' symptoms, and allow essential workers to attend to their duties. Citizens could ask for three types of permits: essential activities, non-essential activities, and a 48-hour permit for specific situations (such as attending a non-urgent medical appointment or aiding relatives). Due to an increase in COVID-19 cases, authorities decided to return to phase 1 on June 29, strengthening mobility restrictions, limiting business hours and imposing more severe fines for no-compliance until July 17 (Alvarez, 2020). Mobility levels decreased, but not as much as at the beginning of the lockdown in late March (La Nación, 2020b). After this period, several commercial, recreational, and cultural activities were enabled, including the opening of parks. By the end of the lockdown period and the beginning of the social distancing phase, mobility had increased significantly from the deep initial drops but was not still near pre-lockdown levels. Appendix Figures A1, A2, and A3 show the evolution of mobility in the city, using data for the number of vehicles counted by government traffic cameras, one-way tickets sold in the public transport system, and Google mobility cellular use data.

In terms of the disease's spread, the Metropolitan Buenos Aires Area initially concentrated most of the country's COVID-19 cases (Costa, 2020). By mid-June, ICU bed occupation exceeded 50 percent (Bär, 2020). The city reached the first peak of contagions on September 15. From that

moment on, there was a slow and steady decline in daily cases through the end of the year (Sigal, 2020). Appendix Figures A4 and A5 present the rates of COVID-19 cases and COVID-19 related deaths for the City of Buenos Aires and Argentina, respectively from March to December 2020.

3. Data and Methodology

3.1. Property Crime and Homicide Datasets

The Ministry of Justice and Security of the City of Buenos Aires produces a rich crime statistics system. These statistics, which are the primary source of criminal information for this study, include reported property crimes and homicides between January 1, 2018, and November 30, 2020, for the City. This period encompasses the mandatory lockdown period (from March 20 to November 9, 2020) and the two previous years, which we use to capture seasonal variations in criminal reports in non-pandemic years.

The criminal reports dataset includes georeferenced individual records for the main property crimes: robberies or thefts (involving or not a weapon), burglaries, and larcenies. Each record also specifies the type of location where the event occurred (home, office, public transportation, street, public space, store, etc.).

The reported homicides dataset provides information on the cause of death, and the time and place of the event (georeferenced). In addition, when known, the dataset includes some sociodemographic characteristics of the victim and the perpetrator.

3.2. Detainees Dataset

We also use data on all detainees in the City between January 1, 2019, and November 30, 2020, provided by the Ministry of Justice and Security of the City of Buenos Aires. The dataset contains information about the type of crime that led to the detention, the crime's date and location, and the detention date. It also specifies the detainee's sociodemographic information, including nationality, sex, age, and place of residence (the neighborhood is specified, but not the exact address). In the dataset, almost half of detainees are residents of the City of Buenos Aires (48.4 percent), 26 percent of detainees live in the Province of Buenos Aires (which surrounds the city), 9.7 percent are homeless, and 0.5 percent have an address in another province or country. There is no residence data for 15.5 percent of detainees.²

3.3. LICIP's Victimization Survey Dataset

² The frequencies of the types of crimes leading to these arrests are 30.1 percent drug trafficking and drug dealing, 23.5 percent robbery, 8.5 percent larcenies, 4.7 percent attack and/or resistance to authority, 4.7 percent injuries, 1.1 percent sexual assaults, 0.7 percent homicides, and 26.7 percent other crimes.

Official crime statistics typically capture a fraction of actual crimes due to underreporting. The gap between actual and reported crime could be particularly relevant when assessing the impact of the COVID-19 pandemic on crime, since the pandemic might have also affected people's ability to report crimes. For our analysis, we complement the official crime statistics with information from a victimization survey produced by the Crime, Institutions, and Policy Research Laboratory of Torcuato Di Tella University (LICIP-UTDT for its initials in Spanish).

Following international standards,³ the survey asks the interviewed subjects whether they or a cohabiting household member has suffered at least one crime in the last 12 months, regardless of whether they reported it to the authorities or not. The survey is conducted monthly by phone to a random and geographically balanced sample of 1,200 households across the country's main urban conglomerates.⁴ It also provides information on respondents' age, sex, and educational level.

We use data from the victimization survey for January 2018 to November 2020. During the period of analysis, 25.5 percent of the interviewed households reported having been victimized. Among the victimized households, the most frequently reported crimes were violent crimes (56.7 percent), larceny of personal belongings (18.8 percent), home burglary (12.6 percent), and car theft (3.7 percent).

3.4. Classification of Areas within the City of Buenos Aires

The 203 km² of the City of Buenos Aires are divided into 15 administrative divisions called communes, which comprise 48 neighborhoods. As shown in Figure 2, the city is separated from the homonymous Province of Buenos Aires by the Riachuelo river (crossed by 6 bridges) and the General Paz Avenue (with 22 vehicle crossings and 26 pedestrian bridges). Due to COVID-19 restrictions, checkpoints were placed at several tolls and highway exits.

³ See United Nations Interregional Crime and Justice Research Institute, ICVS - International Crime Victims Survey (http://www.unicri.it/index.php/services/library_documentation/publications/icvs/data).

⁴ For technical details of the LICIP victimization survey, see the survey's historical reports available in UTDT LICIP's website: <u>https://www.utdt.edu/ver_contenido.php?id_contenido=968&id_item_menu=2156</u> (accessed April 6, 2021). The list of questions is available in the January 2019 report.

Figure 2. The 15 Communes of the City of Buenos Aires and the Entry Checkpoints during the Isolation Period



Source: Government of the City of Buenos Aires (https://data.buenosaires.gob.ar/dataset/puntos-de-acceso-a-laciudad/resource/4b7c5065-2a15-4b5c-bd23-ed3c3046c673, accessed on March 30, 2021).

Crime dynamics might diverge between different areas of the city. To assess if the pandemic had differential effects across types of areas, we used a machine-learning model to classify the location of each reported crime into the following seven land use categories: residential, commercial, business/offices, transportation, parks, industrial and *barrios populares*.⁵ Figure A6 in the Appendix shows how the machine-learning model classified each block and provides further details of the methodology and sources.

3.5. Empirical Strategy

We aim to measure the impact of the COVID-19 pandemic and the subsequent policy responses on crime in the City of Buenos Aires. As discussed above, the pandemic simultaneously affected potential victims, potential criminals, police forces, and the judiciary and penitentiary systems. We take a reduced-form approach to estimate the overall impact of the pandemic on a set of crime and arrest outcomes, as we do not have disaggregated information to disentangle the relative importance of each different channel. We use a linear regression model with time dummies to control for seasonal variations and secular trends in the dependent variables. Formally, we estimate the following models:

$$Y_t = \beta D_t + \gamma_d + \mu_s + \delta_y + \varepsilon_t \quad (1)$$

⁵ The National Ministry of Social Development (Registro Nacional de Barrios Populares) provides maps on all *barrios populares* in Argentina (https://datos.gob.ar/dataset/desarrollo-social-registro-nacional-barrios-populares, accessed on March 30, 2021)

and

$$Y_t = \sum_{i=1}^{8} \beta_i D_{i_t} + \gamma_d + \mu_s + \delta_y + \varepsilon_t \quad (2)$$

where the dependent variable Y_t is a crime or arrest outcome on date t, γ_d is a day-of-the-week dummy variable, μ_s is a week-of-the-year dummy variable, δ_y is a year dummy variable, and ε_t is an idiosyncratic error term.

In equation (1), we consider one dummy variable D_t to indicate the whole lockdown period. In this first model, the coefficient β captures the average impact of the different stages of the lockdown on the outcome variable. In equation (2), the dummy variables D_{it} (i = 1, ..., 8) correspond to each of the seven most relevant stages of the lockdown period, and the last one indicates the beginning of the social distancing phase, as described in Figure 1. These dummy variables start at zero, take the value of one when each stage went into effect, and remain at this value after that. For example, D_1 takes the value of one for all dates since the start of the lockdown (March 20, 2020, onwards) and D_2 takes the value of one for all dates since the start of the second stage of mobility restrictions (April 12, 2020, onwards). Therefore, each β_i coefficient captures the incremental impact of the new stage of the lockdown on the outcome variable (relative to the previous stage). For example, D_2 captures the change in the outcome variable during the second stage of the lockdown's first stage.⁶

The inclusion of the year and week-of-the-year time dummies is critical for our identification strategy and the interpretation of the β coefficients. The year dummy variables capture secular changes in the outcomes. The week-of-the-year dummies capture within-year seasonality. The β coefficients capture the deviation in the outcome variable during the lockdown period relative to its pre-lockdown trend after accounting for typical seasonal changes. Therefore, the β coefficients provide an estimate of the impact of the pandemic (and the different lockdown stages) on criminal activity under the assumption that, if the pandemic had not occurred, the outcome (for example, the number of crimes) would have shown a seasonal evolution as in previous years.⁷ The fact that the timing of the pandemic and the changes in mobility restrictions were unrelated to the crime variables we examine supports the plausibility of this assumption. Moreover, the addition of day-of-the-week dummy variables captures within-week systematic patterns in the outcome variable and helps to increase the precision of the model.

We estimate the models by ordinary least squares (OLS). The time-series nature of our data could lead to autocorrelation in the model residuals and induce bias in the estimation of our standard errors. This bias, in turn, might lead to incorrect statistical inference. We address this issue using

⁶ The change in the outcome variable in the second stage against the pre-lockdown period is given by the sum of D_1 and D_2 .

⁷ This model can be interpreted as a "difference-in-differences" (DiD) model (Leslie and Wilson, 2020). The β coefficient captures the difference in the change in the outcome variable between the periods before and after March 20 in 2020 vs. the change between these two periods in previous years. The two differences occur over the "time" dimension, as opposed to typical DiD models that usually compare pre vs. post changes between different treated and control units. Using this terminology, we can define the year 2020 as our treated unit and the previous years as the control ones.

a heteroscedasticity and autocorrelation consistent (HAC) estimator for the standard errors, following Newey and West (1987). We implement this estimator using the automatic bandwidth selection procedure presented in Newey and West (1994). This estimator allows us to perform (asymptotically) valid statistical inference in the presence of autocorrelation and heteroscedasticity.

To estimate the first model, we use data until October 2020, when the strict lockdown finished. For the second model, we also use the information for November 2020. We estimate these models using the counts of the outcome of interest as the dependent variable. In the case of homicides, we use weekly counts to reduce the number of zero-valued observations. In the analysis of victimization survey data, we use monthly victimization rates. In these last two cases, we use month-of-the-year dummy variables to capture within-year seasonal variations instead of week-of-the-year dummies. This change does not substantially alter the underlying identification assumption or the interpretation of the β coefficient.

For crime and arrests data, we also provide estimates using the inverse hyperbolic sine (IHS) transformation of the daily counts as the dependent variable in the appendix. This transformation, proposed initially by Johnson (1949), is helpful since it approximates the logarithmic function, but, as opposed to the logarithmic function, it is defined at zero (see Burbrige et al., 1988). When using this transformation, the β coefficients can be easily re-expressed to percentage changes.⁸

We complement the previous models with an event-study model. This model allows us to test if, before the pandemic, crime dynamics in 2020 resembled patterns of previous years. The existence of parallel trends in crime dynamics before the pandemic helps validate the counterfactual assumption supporting the causal interpretation of our results. Following Alvarado et al. (2020), we partition our data in fortnights and estimate the following model:

$$Y_t = \sum_{k=-5}^{T} \beta_k D_t^k + \gamma_d + \mu_s + \delta_y + \varepsilon_t \quad (3)$$

In this dynamic version, instead of using a static indicator D_t taking the value of one since the mandatory isolation period began, we include a set of dummy variables D_t^k that take the value of one for fortnight k in the year 2020 and zero otherwise. We set k = 0 for the fortnight starting the week before the beginning of the mandatory lockdown as some changes in mobility (and crime) started to take place once the pandemic was declared, a few days before the mandatory lockdown (see, for example, Appendix Figures A1, A2, and A3). Coefficients between k = -5 and k = -1 capture changes in the outcome variable during the fortnights before the shock. The coefficient for fortnight k = -1 is the omitted category in the estimation and normalized to zero. k = T is the last fortnight of 2020 with available information for each database. If the coefficients for k < -1 show no discernible pattern, it indicates that, before the pandemic, trends for 2020 were "parallel"

⁸ To interpret β as a percentage change, we follow Halvorsen and Palmquist (1980) and use the correction suggested by Kennedy (1981), as explained in Bellemare and Wichman (2019): percent *Change* = $Exp(\beta - 0.5(Var(\beta)) - 1$. We present the β estimate and the percent change in the tables.

to those of the previous years. In turn, the coefficients for $k \ge 0$ capture any deviation in the seasonal fluctuation of crime between the fortnights of 2020 under lockdown and the same fortnights of the previous two years.

Robustness exercises using panel data. To assess the robustness of our main results, we estimate the pandemic's impact on different types of crimes using information disaggregated at the commune and census tract level. For these robustness checks, we use the following panel data model:

$$Y_{c,t} = \beta D_t + \gamma_d + \mu_s + \delta_y + \theta_c + \varepsilon_{c,t} \quad (4)$$

where the dependent variable $Y_{c,t}$ is the value of a crime outcome at date *t* in commune (or census tract) *c*. In this model, we include a set of commune (or census tract) fixed effects (θ_c) to account for the non-varying heterogeneity between these geographical areas. The inclusion of these fixed effects allows us to exploit the within-area variation over time to estimate our coefficient of interest (β). The rest of the variables maintain the same interpretation as in the previous models.

Analysis of detainees' mobility patterns. In addition to estimating the pandemic's impact on victimization rates and the number of different types of crimes and arrests, we examine how the pandemic affected the average distance between the detainees' residence area and the crime location. For this analysis, we use an event level dataset in which each observation corresponds to one arrest and estimate the following model:

$$Y_{a,t} = \beta D_t + \gamma_d + \mu_s + \delta_y + \varepsilon_{a,t} \quad (5)$$

where the dependent variable $Y_{a,t}$ is the distance between the detainee's area of residence and the crime location for arrest *a* occurred on date *t*. The rest of the variables maintain the same interpretation as in the previous models.

4. Results

4.1 Property Crime and Homicides

Figure 3 shows a dramatic fall in (reported) property crime in Buenos Aires after the strict lockdown began on March 20, 2020 (indicated by the dashed vertical line). Before the pandemic, the daily average number of thefts and larcenies was 309. This number fell sharply to 49 on the first week of isolation.⁹ After the large drop, the number of events then recovered gradually to around 185 per day by October 2020, still below the pre-pandemic levels. The drop in property

⁹ Figure 3 also suggests that reported crime started falling shortly before the national lockdown. This premature fall could be the result of an anticipation of individual decisions before the official lockdown once it was well-known that coronavirus cases were already present in the city. It is also plausible that, as some crimes are reported a couple of days after the event, the lockdown could have interrupted those reports.

crime occurred in each of the 15 Buenos Aires communes (see Appendix Figure A7 for each commune).



Figure 3. Daily Number of Reported Thefts and Larcenies in Buenos Aires City

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: 15-day moving average. The dashed line marks the beginning of the mandatory lockdown.

Table 1 presents the results of the estimation of our two main models using the daily number of reports of thefts, burglaries, and larcenies for the whole City of Buenos Aires. The number of daily crime reports fell, on average, by 162 during the strict lockdown period. This fall represents 52.5 percent of the average daily pre-lockdown reports. The number of reports plummeted on the first period of lockdown and then started to recover. In the first stage of the lockdown, the average daily number of reported thefts, burglaries and larcenies fell by 245 (80 percent of the prelockdown average). The posterior evolution of crime reports was sensitive to the intensity of mobility restrictions. The subsequent relaxations in mobility restrictions were accompanied by increases in the number of reports. This steady recovery in the number of crime reports had a short impasse when the city reinstated a strict lockdown between June 29 and July 17, 2020. Table A1 in the appendix reports the results of these estimations using the IHS transformation of the daily count of thefts, burglaries, and larcenies as the dependent variable. Results are robust to this alternative specification of the dependent variable. Results are also robust to alternative model specifications that include fixed effects controlling for the commune of the crime (Appendix Tables A2 and A3), or the census tract where the crime was committed (Appendix Tables A4 and A5).

Figure 4 shows the results of the estimation of the event study model. These results first show that, before the pandemic, crime trends in 2020 were not systematically different from those observed in the previous two years. Moreover, the sharp decline since the mandatory lockdown does not follow a previous pre-lockdown trend.



Figure 4. Event Study of the Impact of the Lockdown on Thefts, Burglaries, and Larcenies

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The event study model includes day of the week, week of the year, and year specific fixed effects. 90 percent confidence intervals reported.

We examine if the change in criminal dynamics varied by the type of area of the city, using the categories described in Section 3.4. Appendix Tables A6 and A7 report the estimations for each type of area separately. We find that theft, burglary, and larceny reports fell across all the different types of areas. The decline was relatively less pronounced in places with a relatively smaller reduction in mobility, such as residential and commercial areas. Business and transportations areas, together with parks, had the largest reductions in theft, burglary, and larceny reports. Figure A8 shows the drop in property crime for each type of area in the City of Buenos Aires.

Interestingly, we did not find a significant effect of the lockdown on property crimes committed inside homes (houses or buildings), but there was more than a 50 percent decline outside home locations (Appendix Table A8).

Crime dynamics in informal settlements (*barrios populares*) during the COVID-19 pandemic deserve special attention. In the City of Buenos Aires, these areas were particularly vulnerable to the pandemic's economic consequences and were the first ones where the local government implemented test-and-trace efforts (the *Detectar* Program). This program's implementation brought a large deployment of public officers and resources to test and trace COVID-19 cases and provide food and shelter to infected people (Suaya and Schargrodsky, 2020). This increased

government presence could have had positive externalities on crime. To estimate the *Detectar* Program's effect on crime, we first run our model on observations for the informal settlements, including an additional dummy variable taking a value of one for the periods after the *Detectar* Program started in each *barrio popular*.¹⁰ Appendix Table A9 reports the results of the estimation of this model. We find that the *Detectar* Program's deployment further reduced theft and larceny reports in these deprived areas.

We also estimate this augmented model using as dependent variable the ratio between the number of reported thefts, burglaries, and larcenies inside *barrios popular i* and the total number of these property crimes reported in the neighborhood where *barrio popular i* is located. This exercise seeks to assess if the evolution of crime in *barrios populares* during the *Detectar* program's implementation was different from that in their surrounding areas. Appendix Table A10 shows the results of the estimation of this model. When considering all *barrios populares*, we find a negative, but not statistically significant (at standard levels) effect of the *Detectar* Program. This estimation includes some *barrios populares* whose neighborhoods were also part of the *Detectar* Program. When we restrict the estimation only to those *barrios populares* for which the program did not cover its surrounding neighborhood, we find a significant negative effect on the ratio of reported property crimes. The increased government presence in response to the health crisis seems to have indirectly reduced crime in these areas.

We also assess if changes in reported thefts, burglaries, and larcenies during the lockdown differed between day and night. Appendix Table A11 presents the results of the estimation of our two main models considering separately reported crimes committed at night (from 10 pm to 6 am)¹¹ or during the day. We find a similar decline in reported thefts and larcenies in the night (-61.3 percent relative to pre-lockdown level) and the day (-51.1 percent).

We then analyze the impact of the lockdown on homicides. The homicide rate in the City of Buenos Aires was 3.32 per 100,000 inhabitants in 2019 (Ministry of Justice and Security of the City of Buenos Aires, 2020). This rate is low compared to the average for Argentina (5.3) and South America (21).¹² Figure 5 shows the 15-day moving average of the number of reported homicides from January 2018 to November 2020. We observe no noticeable change in homicide patterns after the start of the lockdown on March 20, 2020 (vertical dashed line).

¹⁰ The estimated model is: $y_{it} = \alpha L_t + \beta D_{it} + \gamma_d + \mu_s + \delta_y + \varepsilon_{it}$, where y_{it} is the number of thefts, burglaries and larcenies reported in *barrio popular i* during week *t*; L_t is a binary variable that takes 1 since the mandatory lockdown started, and 0 before it was put in place; D_{it} is a binary variable that takes 1 for every week *t* in *barrio popular i* once the *Detectar* Program was rolled out in *barrio popular i* and 0 before; $\gamma_d, \mu_s, \delta_y$ are day-of-the-week, week-of-the-year, and year dummy variables, respectively; and ε_{it} is an idiosyncratic error term.

¹¹ This time interval was chosen because it coincides with the police night shift.

¹² Homicides rate in Argentina and South America refer to 2018. Source: UNODC Data Portal,

https://dataunodc.un.org/content/data/homicide/homicide-rate.

Figure 5. Weekly Number of Reported Homicides in Buenos Aires City



Notes: 15-day moving average. The dashed line marks the beginning of the mandatory lockdown.

We formally assess the lockdown's impact on homicides by estimating the linear regression models described above. Table 2 shows the results of the estimations. We find no significant impact of the lockdown on the number of weekly homicides.¹³ Perhaps this result should not be surprising: our review of the recent literature in the introduction had also found no significant changes in homicides associated with the pandemic. The event study pictured in Figure 6 confirms the lack of effect of the lockdown on the dynamics of homicides in the City of Buenos Aires.

¹³ The coefficient on the mandatory lockdown variable is positive but not statistically significant (at standard confidence levels). Similar results can be observed in Appendix Table A12 using the IHS specification. It draws our attention that there was an increase in homicides when the city went back to phase one of isolation in June/July 2020, later reversed in the final social distancing period.



Figure 6. Event Study of the Impact of the Lockdown on Homicides

Notes: The event study model includes day of the week-, week of the year-, and year-specific fixed effects. 90 percent confidence intervals reported.

We also examine if the homicide dynamics during the lockdown differed between *barrios populares* and other areas of the city. Appendix Table A13 shows the results of the estimations considering homicides inside and outside *barrios populares* separately. We find that the lockdown led to an increase in homicides committed outside *barrios populares* and had no significant impact on homicides inside them. The difference between the impact of the lockdown on homicides outside *barrios populares* is statistically significant (at standard confidence levels).

4.2 Detainees

We assess the impact of the lockdown on police arrests and further investigate its effects on crime dynamics. Figure 7 shows the 15-day moving average of the number of detainees in the City of Buenos Aires from January 2019 to November 2020. The graph shows a large fall in the number of detainees after the start of the lockdown on March 20, 2020 (vertical dashed line). The graph also reveals that the number of detainees had jumped at the beginning of 2020. This initial jump responded to a change in the detention criteria by the local police implemented at the beginning of the year.



Figure 7. Number of Daily Detainees in Buenos Aires City

Notes: 15-day moving average. The dashed line marks the beginning of the mandatory lockdown.

Given these dynamics, in Table 3 we present two alternative estimations for the pandemic's effect on arrests. Models 1 and 2 first consider as counterfactuals that, without the pandemic, the dynamics of detainees during the lockdown would have followed a seasonal pattern as in 2019 starting from the 2020 pre-lockdown levels. The results show a large drop in arrests of about 60 percent. Figure 8 reveals similar findings using the event study specification. Alternatively, in models 3 and 4 of Table 3, we report the results of the estimation of an augmented model, which includes an additional dummy taking the value of one for the pre-pandemic months of 2020 and zero otherwise. In these alternative models, the counterfactual for the 2020 lockdown period is, directly, the same period of 2019. The results are similar, but quantitatively smaller, implying a drop in detainees of about 25 percent.



Figure 8. Event Study of the Impact of the Lockdown on Detainees

Notes: The event study includes day of the week-, week of the year-, and year-specific fixed effects. Additional fixed effects for place of residence of the detainees were included. 90 percent confidence intervals reported.

We estimate the impact of the lockdown on the number of detainees by different types of crime. Table 4 reports the results of these estimations. We find that the lockdown led to significant and large drops in the number of detainees for most crimes, ranging from 30 percent (sexual assaults) to 66.4 percent (thefts using motorbikes) of the pre-lockdown means. The only type of crime that increased during the lockdown was "resistance to authorities." The lockdown led to a sizeable and significant increase in the number of detainees for this type of incident, typically referring to episodes at the checkpoints set up during the lockdown to enforce mobility restrictions. The increase in this type of crime reflects an important consequence of the pandemic on police activity. The police were put in charge of enforcing the mobility restrictions imposed to limit the virus's spread. This responsibility entailed a change in police deployment (from strategies based on crime hot spots to strategies based on transit hot spots) and police interaction with citizens. The increase in "resistance to authorities" episodes reflect the challenges for police officers to enforce new and changing regulations, which some citizens defied. The increase in this type of episode evidences a potential cost and risk of this new role for police agencies. The enforcement of mobility restrictions can generate frictions between citizens and police, which can negatively affect the trust and legitimacy in the police and be counterproductive to its effectiveness in crime prevention and control.

We use the information on detainees to assess if the pandemic and the subsequent mobility restrictions brought a change in the age profile of detainees. School closures and the suspension of social programs reduced access to education and support services among at-risk youth, potentially increasing juvenile delinquency. Table 5 shows the results of the estimation of the impact of the lockdown on the number of detainees, considering detainees under 18 years old,

from 18 to 30 years old, and older than 30 years old, separately. We find a large fall in detainees for the three age groups. The largest drop (relative to the pre-lockdown averages) occurs among juvenile arrests, suggesting that the City of Buenos Aires did not face an increase in juvenile delinquency during the lockdown.

We also use the information on detainees to assess if the lockdown changed criminals' mobility patterns. Every day, the City of Buenos Aires usually receives millions of people from surrounding areas who commute to work, study, do paperwork, receive medical attention, or do recreational activities. The lockdown restricted the entrance to the city only to persons in essential occupations (such as health care workers) or with special authorization. The police enforced these restrictions by controlling all the vehicular accesses to the city (shown in Figure 2). As the dataset includes information on the residence area of detainees and the location of the crime for which they were detained, we can assess if the lockdown controls affected the possibility of outside residents coming to the City of Buenos Aires to commit crimes.

The first three columns of Table 6 show the results of the estimations of the impact of the lockdown on the number of detainees with a residence in the City of Buenos Aires, with a residence in the Province of Buenos Aires (i.e., outside the city), and with no residence (homeless or *situación de calle*), respectively. We find that the impact on the number of detainees is (in relative terms) much larger for detainees with residence outside the city. The lockdown reduced the number of detainees residing in Province of Buenos Aires by 48.4 percent (of the pre-lockdown averages). The impact on the number of detainees with residences with residence in the City of Buenos Aires drop was about half this size: 29.8 percent. These results indicate that mobility restrictions and police checkpoints altered the relative costs and opportunities of committing crimes in the City of Buenos Aires between potential criminals living inside and outside the City.

The lockdown also reduced mobility within the City of Buenos Aires. The police set up some checkpoints inside the city to verify if passersby had the proper authorization to move around. We assess if the mobility restrictions altered criminals' mobility patterns within the city. Table 6 shows the impact of the lockdown on the number of detainees who reside in a different commune where the alleged crime was committed (column 4) and in the same commune (column 5). We find a significant and larger relative decline (46.5 percent, relative to the pre-lockdown mean) in the number of detainees from another commune than in the number of detainees living in the same commune (9 percent, and not statistically significant at standard levels).¹⁴ Our results suggest that mobility restrictions and reinforced street police presence led to relatively fewer detainees from outside the city, and fewer detainees from other communes of the city. During the lockdown, crime became "more local."

We also examine the impact of the lockdown on the distance from the location of the crime to the residence area of the detainee. Appendix Table A15 shows that the lockdown led to a significant and large distance reduction of more than one mile when considering all detainees (column 1),

¹⁴ We also estimate these models using the IHS transformation of the daily counts as the dependent variable. Results, reported in Appendix Table A14, show similar patterns, but the reduction in the number of detainees from the same commune is significant under this specification.

but smaller and less precise when restricting the analysis to detainees who reside in the City of Buenos Aires (column 2). The rest of this table performs the same analysis considering separately different types of crime. We find that the lockdown reduced the average distance between the crime location and the detainees' area of residence for most of them.

4.3 Victimization Surveys

The previous results rely on official crime reports. Official statistics are an extremely useful source of information, as they provide many details on the characteristics and dynamics of criminal activity. However, examining these statistics might not be enough to assess the impact of the pandemic on crime. The pandemic and the subsequent containment measures may have also affected crime reporting rates. If this is the case, the previous analysis might be confounding actual crime reductions with decreases in crime reporting.

We assess the robustness of our results using information from the LICIP's monthly victimization survey described in section 3.3 above. Figure 9 presents the January 2018-November 2020 evolution of crime victimization for the City of Buenos Aires, the suburban Greater Buenos Aires area (i.e., municipalities in the Province of Buenos closest to the City), and the rest of the country. There is a large drop in reported victimization in the three areas right after the implementation of the lockdown. Indeed, April 2020 showed the lowest national victimization rate measured by this survey since its first round (March 2008). In May 2020, the survey registered the lowest victimization rate for the City of Buenos Aires when it reached 10 percent, well below its prepandemic levels of around 25 percent. The victimization rate began to increase after that initial fall, but it remained below previous years' levels.



Figure 9. Percentage of Victimized Interviewed Households in Buenos Aires City, Greater Buenos Aires and Rest of the country

Source: Authors' calculations based on data from the Crime, Institutions and Policy Research Laboratory of Torcuato Di Tella University (LICIP-UTDT).

We also estimate the impact of the pandemic and the lockdown on these monthly victimization data. Table 7 reports the results of these estimations for the City of Buenos Aires in the first two columns. We find that the pandemic and the lockdown led to a large and statistically significant fall in the reported victimization rate of 5.5 percentage points, a 20.7 percent drop relative to the pre-lockdown mean. As with official crime statistics, we observe a large initial fall and a posterior recovery. Note that the relative magnitudes in the falls in reported victimization and official crime reports are hard to compare because the survey victimization question refers to incidents in the last 12 months (and not solely to the last month). The reductions in reported victimization confirm that the observed falls in reported crime during the lockdown are not only an artifact of increased underreporting.

We also use the information from the victimization survey to scrutinize if the changes in criminal mobility patterns induced by the lockdown and the mobility controls led to crime displacement. We seek to assess if increased controls in the access to the City of Buenos Aires, which reduced the share of detainees from outside the City, led to a relative increase in crime in its surrounding areas. We take advantage of the fact that the survey provides information on households in the Greater Buenos Aires area and the other main urban conglomerates in the country. In the rest of Table 7, we separately estimate the impact of the lockdown on the reported victimization for households living in the Greater Buenos Aires area and the other main urban conglomerates of the Argentine urban

conglomerates. Although the pandemic's effects are somewhat weaker in these regions, Table 8 shows that the lockdown's impacts on reported victimization in the Greater Buenos Aires and the rest of the country are not statistically different from that for the City of Buenos Aires. This relatively homogeneous drop in crime across the City of Buenos Aires, its surrounding areas, and the whole country suggests that the reduced crime in the City was not displaced to other areas.

5. Conclusions

This paper studies the impact of the COVID-19 pandemic and the subsequent lockdown on criminal activity in the City of Buenos Aires, Argentina. We find a large and significant decline in criminal activity during the lockdown, from March to November 2020. The lockdown led to a large drop in property crimes reported to official agencies (-52.5 percent), arrests made by the police (-59.3 percent), and crime reported in victimization surveys (-20.7 percent), but not in homicides. Moreover, the fall in criminal activity was widespread across different areas of the city.

Our paper contributes to the literature on the crime impact of the COVID-19 pandemic in two main ways. First, we provide information on the evolution of crime during the pandemic in a large Latin American capital city, a relatively understudied setting. Second, by combining information from both official reports and a victimization survey, our results suggest that the observed fall in reported crime during the lockdown was not an artifact of increased underreporting. Most papers that examine the pandemic's impact on crime rely solely on official statistics and can, therefore, confound actual crime changes with changes in crime reporting. Combining these different data sources is critical to accurately assess the impact of shocks that might simultaneously affect crime and reporting rates. This observation applies to the pandemic but extends to other changes in policing strategies or in other determinants of crime (especially in contexts with high rates of crime underreporting).

Moreover, we contribute to the analysis of crime displacement by assessing the lockdown's impact on criminals' mobility patterns. The pandemic led to reduced mobility and strict controls in the accesses to the City of Buenos Aires. These changes likely increased the cost of committing crimes inside the city for potential criminals residing in the suburban metropolitan area. We find that the lockdown led to relatively fewer detainees from outside the city and from other communes of the city. In short, we find that during the lockdown crime became "more local". But we find no evidence that this reduction in the participation of detainees from outside the City of Buenos Aires led to a displacement of crime to suburban areas. These results align with the hypothesis that focalized place-based interventions have the potential to reduce overall crime rates. The pandemic, however, is an exceptional event, and the enforcement of mobility restrictions was extraordinarily strong during the lockdown. We should be cautious about the external validity of these displacement results, as local interventions may not have similar overall effects in other contexts.

Our results provide some additional insights for the design of citizen security policies. We show that not all crimes are the same. The stark difference between homicide dynamics and property

crimes dynamics during the lockdown illustrates how these different crimes have distinct drivers. These heterogeneous dynamics highlight the need to design specific strategies to address different crimes, especially homicide -the most violent and costly one. Moreover, the data also highlight the inter-jurisdictional nature of crime, suggesting the need to strengthen judicial cooperation between jurisdictions in criminal prosecution.

We also document a significant increase in arrests for "resistance to authorities", which mainly included conflicts at the checkpoints set up during the lockdown to enforce the mobility restrictions. These episodes reflect new challenges faced by police agencies worldwide. Suddenly, police officers were required to enforce new and changing mobility restrictions in several countries. The increase in these types of detentions in the City of Buenos Aires illustrates the potential costs and risks of this new role for police agencies. The enforcement of mobility restrictions can cause frictions between citizens and police, negatively affecting police's trustworthiness and legitimacy, and limiting its effectiveness in crime prevention and control. It is critical that authorities carefully evaluate who will enforce mobility restrictions and how to do so, considering the need to provide training for police officers or create specific agencies to this end.

Finally, we find that the increased government presence for testing and tracing COVID-19 cases in informal settlements appears to have led to an additional decrease in crime in these areas beyond the overall impact of the pandemic. This result exemplifies how increased and targeted government interventions for the provision of social services can have positive externalities and complement law enforcement agencies' efforts in the fight against crime.

The available data from the Buenos Aires City government, covering up to November 2020 at the time of writing, allow us to study the pandemic's short-term impact on crime. But the pandemic is not at all over and, in response to a second and stronger COVID-19 wave, lockdown restrictions were reimposed in April 2021. The last available LICIP victimization survey shows that, relative to a 12.9 percent average victimization rate during the most stringent lockdown period of April-May 2020, crime victimization has already rebounded to 22.8 percent in April-May 2021, but it is still below the pre-pandemic level of 29.6 percent victimization rate of April-May 2019. The recency of this massive shock, and its potentially lasting effects on unemployment, poverty, and inequality, prevent us from speculating on its longer-term impact. Moreover, Latin America, already a region with a high level of inequality and crime, has been hit particularly hard by the pandemic, both in health and economic terms. The overall long-term impacts of the COVID-19 pandemic on crime are highly uncertain and call for future close monitoring of crime dynamics.

References

- Abt, T., Rosenfeld, R., and Lopez, E. (2020). COVID-19 and Homicide: Final Report to Arnold Ventures. Arnold Ventures .
- ACLU Analytics. (2020, July 27). Decarceration and Crime During COVID-19. ACLU.
- Alvarado, N., Norza, E., Perez-Vincent, S. M., Tobón, S., and Vanegas-Arias, M. (2020, October). Evolución de la seguridad ciudadana en Colombia en tiempos del COVID-19. Banco Interamericano de Desarrollo.
- Alvarado, N., Sutton, H., and Laborda, L. (2020). COVID-19 and Police Agency Operations in Latin America and the Caribbean . Washington, DC : Inter-American Development Bank.
- Alvarez, J. (2020, June 26). Cuarentena a fase 1: "El AMBA está contagiando al resto del país y tenemos que frenarlo", dijo Fernández . *La Voz*.
- Andresen, M.A., Hodgkinson, T. (2020, November) Somehow I always end up alone: COVID-19, social isolation and crime in Queensland, Australia. *Crime Science*, 9(25).
- Ashby, M. P. (2020, May). Initial evidence on the relationship between the coronavirus pandemic and crime in the United States. *Crime Science*, 9(6).
- Balmori de la Miyar, J. R., Hoehn-Velasco, L., and Silverio-Murillo, A. (2020, September). Druglords don't stay at home: COVID-19 pandemic and crime patterns in Mexico City. *Journal of Criminal Justice*.
- Bär, N. (2020, June 16). Coronavirus en la Argentina. Preocupa la ocupación de las terapias intensivas en la ciudad y el conurbano bonaerense . *La Nación*.
- Bellemare, M. F., & Wichman, C. J. (2020). Elasticities and the inverse hyperbolic sine transformation. *Oxford Bulletin of Economics and Statistics*, 82(1), 50–61.
- Blattman, Christopher, Donald Green, Daniel Ortega and Santiago Tobon (2021). "Placed Intervention at scale: the direct and spillover effects of policing and city services on crime." *Journal of the European Economic Association*, forthcoming.
- Braga, Anthony A., Andrew V. Papachristos, and David M. Hureau (2014). "The effects of hot spots policing on crime: An updated systematic review and meta-analysis." *Justice Quarterly*, 31(4), 633–663.
- Braga, Anthony A., Brandon Turchan, Andrew V. Papachristos, and David M. Hureau (2019). "Hot spots policing of small geographic areas effects on crime." *Campbell Systematic Reviews*, 15(3), e1046.
- Boxall, H., Morgan, A., & Brown, R. (2020). The prevalence of domestic violence among women during the COVID-19 pandemic. *Australasian Policing*, 12(3), 38–46.
- Burbidge, J. B., Magee, L., and Robb, A. L. (1988). Alternative Transformations to Handle Extreme Values of the Dependent Variable. *Journal of the American Statistical Association*, 83(401), 123-127.
- Calderon-Anyosa, R. J., and Kaufman, J. S. (2020, July). Impact of COVID-19 Lockdown Policy on Homicide, Suicide, and Motor Vehicle Deaths in Peru.
- Campedelli, G. M., Aziani, A., and Favarin, S. (2020). Exploring the Effect of 2019n-CoV Containment Policies on Crime: The Case of Los Angeles.
- Collazos, Daniela, Eduardo Garcia, Daniel Mejia, Daniel Ortega, and Santiago Tobon (2020). "Hotspots policing in a high crime environment: An experimental evaluation in Medellin." *Journal of Experimental Criminology*, forthcoming.
- Costa, J. M. (2020, May 10). Mapa del Covid-19: el 66 percent de los infectados de todo el país viven en la Capital y el conurbano. *La Nación*.
- Di Tella, Rafael and Ernesto Schargrodsky (2004). "Do Police Reduce Crime? Estimates Using the Allocation of Police Forces After a Terrorist Attack." *American Economic Review*, 94(1), 115–33.

- Draca, Mirko, Stephen Machin, and Robert Witt (2011). "Panic on the streets of london: Police, crime, and the july 2005 terror attacks." *American Economic Review*, 101(5), 2157–181.
- Fajnzylber, Pablo, Daniel Lederman, and Norman Loayza (2002). "Inequality and Violent Crime," The Journal of Law and Economics 45, no. 1: pp. 1–39.
- Felson, M., Jiang, S., and Xu, Y. (2020, June). Routine activity effects of the Covid-19 pandemic on burglary in Detroit, March, 2020. *Crime Science*, *9*(10).
- GCBA (2020a, May 9). *Las medidas para la nueva fase de la cuarentena.* Obtenido de https://www.buenosaires.gob.ar/node/88275/noticias/coronavirus-instant-articlemedidas-nueva-fase-cuarentena
- GCBA (2020b, June 5). *Nueva etapa: se agregan medidas de bienestar integral para las personas en la Ciudad.* Obtenido de https://www.buenosaires.gob.ar/jefedegobierno/noticias/rodriguez-larretala-intencion-esir-una-etapa-donde-podamos-contemplar-el
- GCBA (2020c, July 21). El Plan Integral y Gradual de Puesta en Marcha de la Ciudad. Retrieved from https://www.buenosaires.gob.ar/coronavirus/noticias/cuarentena-planintegral-gradual-aperturas-por-dia-en-la-primera
- GCBA (2020d, November 9). Las medidas vigentes desde este lunes 9 de noviembre . Retrieved from https://www.buenosaires.gob.ar/laciudad/noticias/nuevas-medidas-yaperturas-del-plan-de-puesta-en-marcha-de-la-ciudad
- Gerell, M., Kardell, J., and Kindgren, J. (2020, October 11). Minor covid-19 association with crime in Sweden, a ten week follow up. *Crime Science*, *9*(19).
- Google LLC. (n.d.). *Google COVID-19 Community Mobility Reports*". Retrieved December 28, 2020, from https://www.google.com/covid19/mobility/
- Hale, T., Boby, T., Angrist, N., Cameron-Blake, E., Hallas, L., Kira, B., . . . Webster, S. (2020, December 10). Variation in Government Responses to COVID- 19 Version 9.0. *Blavatnik School of Government Working Paper*.
- Halford, E., Dixon, A., Farrell, G., Malleson, N., and Tilley, N. (2020, July). Crime and coronavirus: social distancing, lockdown, and the mobility elasticity of crime. *Crime Science*, *9*(11).
- Halvorsen, R., & Palmquist, R. (1980). The interpretation of dummy variables in semilogarithmic equations. *American Economic Review*, 70(3), 474-475.
- Hodgkinson, T., and Andresen, M. A. (2020, July-August). Show me a man or a woman alone and I'll show you a saint: Changes in the frequency of criminal incidents during the COVID-19 pandemic. *Journal of Criminal Justice*, 69.
- Hodgkinson, Tarah, Gregory Saville, Martin A Andresen (2020). "The Diffusion of Detriment: Tracking Displacement Using a City-Wide Mixed Methods Approach," *The British Journal of Criminology*, Volume 60, Issue 1, January, pp. 198– 218, https://doi.org/10.1093/bjc/azz025
- Johnson, N. L. (1949). Systems of Frequency Curves Generated by Methods of Translation. *Biometrika, 36*, 149–76.
- Kennedy, P. E. (1981). Estimation with correctly interpreted dummy variables in semilogarithmic equations [the interpretation of dummy variables in semilogarithmic equations]. *American Economic Review*, 71(4), 801–01.
- La Nación. (2020a, May 8). Coronavirus en la Argentina: se duplicaron los usuarios que utilizaron el transporte público en el AMBA. *La Nación*.
- La Nación. (2020b, July 2). Índice de movilidad: menos gente en la calle, pero más que al inicio de la cuarentena, según la Universidad de San Martín y Movistar. *La Nación*.
- Leslie, E., and Wilson, R. (2020). Sheltering in place and domestic violence: Evidence from calls for service during COVID-19. *Journal of Public Economics*, 189, 104241.
- Londoño, E., Andreoni, M., and Casado, L. (2020, April 28). El coronavirus ataca las cárceles y cientos de miles de presos son liberados. *The New York Times*.

Lum, C., Maupin, C., and Stoltz, M. (2020). "The impact of COVID-19 on law enforcement agencies." International Association of Chiefs of Police and the Center for Evidence-Based Crime Policy, George Mason University.

Ministerio de Salud de la Ciudad de Buenos Aires: Dirección Nacional de Epidemiología y Análisis de la Situación de Salud, Área de Vigilancia. *COVID-19. Casos registrados en la República Argentina*. Retrieved on January 31, 2021, from http://datos.salud.gob.ar/dataset/covid-19-casos-registrados-en-la-republica-argentina

Ministry of Justice and Security of the City of Buenos Aires (2020). *Informe de Estadística Criminal Ciudad Autónoma de Buenos Aires: Años 2018 y 2019.* Ciudad Autónoma de Buenos Aires.

- Mohler, G. O. (2020, June). Impact of social distancing during COVID-19 pandemic on crime in Los Angeles and Indianapolis. *Journal of Criminal Justice, 68*.
- Newey W. K. and West K. D. (1987). "A Simple, Positive Semi-Definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix." *Econometrica*, 55, 703–08.
- Newey W. K. and West K. D. (1994). "Automatic Lag Selection in Covariance Matrix Estimation." *Review of Economic Studies*, 61, 631–53.
- Nivette, A. E., Zahnow, R., Aguilar, R. et al. (2021) "A global analysis of the impact of COVID-19 stay-at-home restrictions on crime." *Nature Human Behavior*. https://doi.org/10.1038/s41562-021-01139-z
- Payne, J. L., and Morgan, A. (2020). Property crime during the COVID-19 pandemic: a comparison of recorded offence rates and dynamic forecasts (ARIMA) for March 2020 in Queensland, Australia.
- Perez-Vincent, S. M., Carreras, E., Gibbons, M. A., Murphy, T. E., and Rossi, M. A. (2020, July). COVID-19 Lockdowns and Domestic Violence: Evidence from Two Studies in Argentina. Inter-American Development Bank.
- Rosenfeld, R., and Lopez, E. (2020, July). Pandemic, Social Unrest, and Crime in U.S. Cities. *Council on Criminal Justice*.
- Raphael, Steven, and Rudolf Winter-Ebmer (2001). "Identifying the Effect of Unemployment on Crime." *The Journal of Law & Economics*, vol. 44, no. 1, pp. 259–83.
- Schargrodsky, Ernesto, and Lucía Freira (2021). "Inequality and Crime in Latin America and the Caribbean: New Data for an Old Question", mimeo, UNDP.
- Semple, K., and Ahmed, A. (2020, April 13). El virus disminuye la criminalidad en América Latina (por ahora). *The New York Times*.
- Shayegh, S., and Malpede, M. (2020, April). "Staying Home Saves Lives, Really!," SSRN Electronic Journal 10.2139/ssrn.3567394
- Sherman, Lawrence, Stephen Williams, Ariel Barak, Lucinda R. Strang, Neil Wain, Molly Slothower, and Andre Norton (2014). "An Integrated Theory of Hot Spots Patrol Strategy: Implementing Prevention by Scaling Up and Feeding Back." *Journal of Contemporary Criminal Justice*, 30(2), 95–122.
- Sigal, P. (2020, October 6). Coronavirus: la matemática confirma que la provincia de Buenos Aires también superó el pico de casos. *Clarín*.
- Soares, R. R. (2004, July). "Crime Reporting as a Measure of Institutional Development." *Economic Development and Cultural Change*, 52(4), 851–71.
- Suaya, Agustina, and Schargrodsky, Ernesto (2020, November). "Estrategia de contención del COVID-19 en el Barrio Padre Carlos Mugica. Evaluación de impacto." Buenos Aires: CIPPEC.
- UNDP. (2019). *Human Development Report 2019*. New York: United Nations Development Programme.
- United Nations Office on Drugs and Crime. (2020). *The impact of COVID-19 on organized crime.* Vienna: UNODC.

Weisburd, David, D. Groff, and S. M. Yang (2012). *The Criminology of Place: Street Segments and Our Understanding of the Crime Problem*. Oxford University Press, New York.

- Weisburd, David and Cody Telep (2016). "Hot Spots Policing: What We Know and What We Need to Know." *Journal of Experimental Criminology*, 30(2), 200–20.
- World Bank. (2020, June 8). *Global Economic Prospects, June 2020.* Washington, DC: World Bank.
- World Health Organization. (2020). WHO Director-General's opening remarks at the media briefing on COVID-19 March 11, 2020.

		
	Model 1	Model 2
Mandatory lockdown (-)	-161.978***	-244.931***
	(10.371)	(11.813)
Authorization of essential workers (+)		35.422***
		(9.637)
Opening of local shops (+)		26.424***
		(8.712)
Opening of shops, permit for runners and		46.798***
recreational outings for children on weekends (+)		(7.910)
Return to phase 1 (strict lockdown) (-)		-20.989**
		(9.658)
Staggered reopening of the city (+)		18.543**
		(8.310)
Opening of bars and outdoors social gatherings		20.398**
(+)		(8.625)
Social distancing phase (+)		22.820 [*]
		(12.207)
Pre-lockdown mean	308.655	308.655
Change with respect to pre-lockdown levels	-52.5%	-
Year DV	YES	YES
Week DV	YES	YES
Day of the week DV	YES	YES
R ²	0.813	0.846
Adj. R ²	0.799	0.834
Num. obs.	1035	1065

Table 1: Estimation of the Effect of the Lockdown on the Number of Daily Reports forThefts, Burglaries and Larcenies in the City of Buenos Aires

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the daily number of reports of thefts, burglaries, and larcenies. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table 2: Estimation of the Effect of the Lockdown on the Number of Weekly Reports of
Homicides in the City of Buenos Aires

	Model 1	Model 2
Mandatory lockdown (-)	0.375	-0.421
	(0.588)	(0.861)
Authorization of essential workers (+)		0.979
		(0.763)
Opening of local shops (+)		0.003
		(0.948)
Opening of shops, permit for runners and recreational outings for		-1.128
children on weekends (+)		(0.846)
Return to phase 1 (strict lockdown) (-)		2.788***
		(0.689)
Staggered reopening of the city (+)		-0.103
		(1.156)
Opening of bars and outdoor social gatherings (+)		-2.312 [*]
		(1.182)
Pre-lockdown mean	2.278	2.278
Change with respect to pre-lockdown mean	16.45%	
Year DV	YES	YES
Month DV	YES	YES
R ²	0.122	0.187
Adj. R ²	0.035	0.066
Num. obs.	156	156

Notes: The dependent variable is the weekly number of homicide reports. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Models include year- and month-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Model 1	Model 2	Model 3	Model 4
Mandatory lockdown (-)	-51.545***	-38.867***	-20.851***	-7.907**
	(4.204)	(5.084)	(3.508)	(4.009)
Authorization of essential workers (+)		-5.557		-5.557
		(5.287)		(4.840)
Opening of local shops (+)		3.214		3.214
		(4.450)		(3.430)
Opening of shops, permit for runners and		9.298**		9.298***
recreational outings for children on weekends (+)		(4.544)		(3.204)
Return to phase 1 (strict lockdown) (-)		-21.416***		-21.416***
		(4.035)		(3.109)
Staggered reopening of the city (+)		8.937***		8.937***
		(2.909)		(2.226)
Opening of bars and outdoor social gatherings (+)		-6.625		-6.625
		(5.744)		(5.953)
Social distancing phase (+)		-71.436***		-71.436***
		(5.882)		(6.201)
2020 pre-lockdown		. ,	30.694***	30.959***
			(2.558)	(2.161)
Pre-lockdown mean	87.108	87.108	87.108	87.108
Lockdown chg. with respect to pre-lockdown mean	-59.2%	-	-23.94%	-
2019 mean	82.204	82.204	82.204	82.204
Lockdown chg. with respect to 2019 mean	-62.7%	-	-25.4%	-
Year DV	YES	YES	NO	NO
Week DV	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES
R ²	0.577	0.686	0.561	0.680
Adj. R ²	0.530	0.647	0.521	0.647

Table 3: Estimation of the Effect of the Lockdown on the Number of Daily Detainees in the City of Buenos Aires

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Num. obs.

700

700

700

700

Notes: The dependent variable is the daily number of detainees. All four models include all data between January 1, 2019 and November 30, 2020. Models 3 and 4 further introduce a dummy variable that takes 1 only for the observations in 2020 before the introduction of the mandatory lockdown. Models 1 and 2 include year-, week- and day of the weekspecific dummy variables (DV) and Models 3 and 4 include week- and day of the week-specific DV. HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

i ypc									
	Drug trafficking	Theft	Thieves on Motorbike	Larceny	Drug dealing	Sexual assault	Homicide	Resistance to authorities	Others
Average effect	-16.515***	-9.532***	-0.642***	-6.161***	-3.009***	-0.409**	-0.290**	2.184***	-5.859***
of lockdown	(2.000)	(0.866)	(0.155)	(0.487)	(0.503)	(0.166)	(0.127)	(0.820)	(1.820)
Mean Jan/Feb 2020	28.55	26.2	0.967	11.03	4.8	1.35	0.65	2.583	21.933
Change to Jan/Feb 2020	-57.85%	-36.3%	-66.39%	-55.8%	-62.6%	-30.3%	-44.6%	84.55%	-26.7%
Pre-lockdown mean	23.158	19.707	0.872	7.622	2.86	1.043	0.5	4.775	21.232
Change to pre- lockdown levels	-71.31%	-48.3%	-73.62%	-80.8%	-105.%	-39.2%	-58%	45.74%	-27.5%
Year DV	YES	YES	YES	YES	YES	YES	YES	YES	YES
Week DV	YES	YES	YES	YES	YES	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES	YES	YES	YES	YES	YES
R ²	0.606	0.318	0.158	0.333	0.231	0.141	0.135	0.335	0.323
Adj. R ²	0.561	0.241	0.063	0.258	0.144	0.043	0.037	0.260	0.247
Num. obs.	700	700	700	700	700	700	700	700	700

Table 4: Estimation of the Effect of the Lockdown on the Number of Daily Detainees in the City of Buenos Aires by Crime Type

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina). Notes: The dependent variable is the number of detainees per day for each type of crime. All models consider the period January 2019 through October 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	<18	18-30	>30		
Average effect of lockdown	-5.980***	-25.281***	-15.066***		
	(0.382)	(1.555)	(1.452)		
Mean Jan/Feb 2020	8.717	53.65	43.85		
Change with respect to Jan/Feb 2020	-68.6%	-47.1%	-34.4%		
Pre-lockdown mean	5.75	44.953	36.419		
Change with respect to pre-lockdown levels	-104%	-56.2%	-41.4%		
Year DV	YES	YES	YES		
Week DV	YES	YES	YES		
Day of the week DV	YES	YES	YES		
R^2	0.334	0.533	0.460		
Adj. R ²	0.259	0.480	0.399		
Num. obs.	700	700	700		

Table 5: Estimation of the Effect of the Lockdown on the Number of Daily Detainees in the City of Buenos Aires by Age

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the number of detainees each day. The first model includes detainees under 18 years old. The second model includes detainees between 18 and 30 years old. The third model includes detainees over 30 years old. All models consider the period January 2019 through October 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	City of Buenos Aires	Province of Buenos Aires	Homeless	Different commune	Same commune
Mandatory lockdown	-12.098***	-12.413***	-2.369***	-9.703***	-1.506
	(1.214)	(1.461)	(0.520)	(0.934)	(0.948)
Mean Jan/Feb 2020	45.933	26.083	13.150	33.600	10.967
Change with respect to Jan/Feb 2020	-26.3%	-47.6%	-18%	-28.9%	-13.7%
Pre-lockdown mean	40.655	25.644	6.529	20.863	16.782
Change with respect to pre- lockdown levels	-29.8%	-48.4%	-36.3%	-46.5%	-9%
Year DV	YES	YES	YES	YES	YES
Week DV	YES	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES	YES
R ²	0.383	0.618	0.531	0.543	0.465
Adj. R ²	0.311	0.574	0.477	0.490	0.403
Num. obs.	670	670	670	670	670

Table 6: Estimation of the Effect of the Lockdown on the Number of Daily Detainees in
the City of Buenos Aires by Address of the Detainee

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the number of detainees each day. The first model only includes detainees with an address in the City of Buenos Aires; the second model includes only detainees with an address in the Province of Buenos Aires; the third model includes only homeless detainees; the fourth and fifth models consider detainees with an address in a different commune or the same commune of the City of Buenos Aires, respectively. All models consider the period January 2019 through October 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	City of Buenos Aires		Greater Buenos Aires		Rest of the Country	
	Model 1	Model 2	Model 1	Model 2	Model 1	Model 2
Mandatory lockdown	-0.055***	-0.124***	-0.023	-0.164***	-0.023	-0.096***
	(0.018)	(0.010)	(0.024)	(0.007)	(0.025)	(0.008)
Mandatory lockdown		-0.015		0.121***		0.018
+ 1 month		(0.020)		(0.017)		(0.028)
Mandatory lockdown		0.094		-0.001		-0.014
+ 2 months		(0.093)		(0.026)		(0.033)
Mandatory lockdown		-0.006		-0.022		0.106**
+ 3 months		(0.021)		(0.034)		(0.038)
Mandatory lockdown		-0.004		0.087		0.045*
+ 4 months		(0.019)		(0.050)		(0.025)
Mandatory lockdown		0.233***		0.014		-0.050***
+ 5 months		(0.010)		(0.015)		(0.004)
Mandatory lockdown		-0.309***		-0.000		-0.032
+ 6 months		(0.007)		(0.005)		(0.023)
Mandatory lockdown		0.059**		0.010		0.046*
+ 7 months		(0.022)		(0.016)		(0.023)
Pre-lockdown mean	0.266	0.266	0.316	0.316	0.287	0.287
Change with respect to pre-lockdown levels	-20.7%	-	-7.3%	-	-8%	-
Year DV	YES	YES	YES	YES	YES	YES
Month DV	YES	YES	YES	YES	YES	YES
Num. obs.	35	35	35	35	35	35
R ²	0.445	0.804	0.554	0.874	0.421	0.747
Adj. R ²	0.142	0.555	0.311	0.714	0.105	0.426

Table 7: Estimation of the Effect of the Lockdown on the Percentage of VictimizedHouseholds

Source: Authors' calculations based on data from the Crime, Institutions and Policy Research Laboratory of Torcuato Di Tella University (LICIP-UTDT).

Notes: the dependent variable is the proportion of surveyed individuals who reported having been victims of a crime in the preceding 12 months to the survey. All models consider monthly data for the period January 2019 through November 2020. The mandatory lockdown begins in April 2020. Models include year- and month-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Model 1
Lockdown	-0.054**
	(0.022)
Lockdown * GBA	0.019
	(0.026)
Lockdown * Rest of the country	0.042
	(0.033)
Year FE	YES
Month FE	YES
Region FE	YES
F-stat (proj. model)	1.840
F-test p-value (proj. model)	0.146
R ²	0.472
Adj. R ²	0.361
Num. obs.	105

Table 8: Differences in the Impact on Victimization Across Regions

Source: Authors' calculations based on data from the Crime, Institutions and Policy Research Laboratory of Torcuato Di Tella University (LICIP-UTDT).

Notes: We omit the coefficient corresponding to the City of Buenos Aires. Therefore, and because we are including region-specific fixed effects, the Lockdown coefficient captures the isolated impact on the City of Buenos Aires. Since neither the interaction with GBA nor with the rest of the country are statistically significant, we cannot reject that the results for those places are similar from the ones obtained for the City of Buenos Aires. In addition, coefficients for GBA and for the rest of the country are not statistically different of each other. HAC standard errors in parentheses. ***p < 0.01; **p < 0.05; *p < 0.1.

Online Appendix

Table A1: Estimation of the Effect of the Lockdown on the Number of Daily Reports forThefts, Burglaries and Larcenies in the City of Buenos Aires (IHS Transformation)

	Model 1	Model 2
Mandatory lockdown (-)	-0.878***	-1.734***
	(0.073)	(0.048)
Authorization of essential workers (+)		0.451***
		(0.055)
Opening of local shops (+)		0.303***
		(0.056)
Opening of shops, permit for runners		0.266***
and recreational outings for children on		(0.038)
Return to phase 1 (strict lockdown) (-)		-0.125
		(0.067)
Staggered reopening of the city (+)		0.179
		(0.057)
Opening of bars and outdoors social		0.171***
gatherings (+)		(0.050)
Social distancing phase (+)		0.124**
		(0.061)
Chg in % terms	-58.4%	-
Year DV	YES	YES
Week DV	YES	YES
Day of the week DV	YES	YES
R ²	0.758	0.852
Adj. R ²	0.740	0.842
Num. obs.	1035	1065

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the IHS transformation of the daily number of reports of thefts, burglaries and larcenies. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Models include year, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Model 1	Model 2
Mandatory lockdown (-)	-9.726***	-15.214***
	(0.347)	(0.512)
Authorization of essential workers (+)		2.225***
		(0.557)
Opening of local shops (+)		1.352**
		(0.536)
Opening of shops, permit for runners and recreational outings for		3.113***
children on weekends (+)		(0.590)
Return to phase 1 (strict lockdown) (-)		-1.446**
		(0.643)
Staggered reopening of the city (+)		1.227**
		(0.502)
Opening of bars and outdoors social gatherings (+)		1.194***
		(0.389)
Social distancing phase (+)		1.395**
		(0.591)
Pre-lockdown mean	19.030	19.030
Change with respect to pre-lockdown levels	-51.11%	-
Year FE	YES	YES
Week FE	YES	YES
Day of the week FE	YES	YES
Commune FE	YES	YES
R ²	0.205	0.213
Adj. R ²	0.201	0.210
Num. obs.	16010	16468

Table A2: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires at the Commune Level (in Absolute Levels)

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the daily number of reports of thefts, burglaries and larcenies. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Clustered standard errors by day in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A3: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires at the Commune Level (IHS Transformation)

	Model 1	Model 2
Mandatory lockdown (-)	-0.783***	-1.739***
	(0.029)	(0.043)
Authorization of essential workers (+)		0.462***
		(0.051)
Opening of local shops (+)		0.302***
		(0.047)
Opening of shops, permit for runners and recreational outings for		0.300***
children on weekends (+)		(0.052)
Return to phase 1 (strict lockdown) (-)		-0.128**
		(0.055)
Staggered reopening of the city (+)		0.205***
		(0.041)
Opening of bars and outdoor social gatherings (+)		0.169***
		(0.036)
Social distancing phase (+)		0.122***
		(0.045)
Change in % terms	-54.3%	-
Year FE	YES	YES
Week FE	YES	YES
Day of the week FE	YES	YES
Commune FE	YES	YES
R ²	0.846	0.872
Adj. R ²	0.845	0.871
Num. obs.	17040	17040

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the IHS transformation of the daily number of reports of thefts, burglaries and larcenies per commune. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Clustered standard errors by day in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A4: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires at the Census Tract Level (in Absolute Levels)

	Model 1	Model 2
Mandatory lockdown (-)	-0.044***	-0.069***
	(0.002)	(0.002)
Authorization of essential workers (+)		0.010***
		(0.003)
Opening of local shops (+)		0.006**
		(0.002)
Opening of shops, permit for runners and recreational outings for		0.014***
children on weekends (+)		(0.003)
Return to phase 1 (strict lockdown) (-)		-0.006**
		(0.003)
Staggered reopening of the city (+)		0.006**
		(0.002)
Opening of bars and outdoor social gatherings (+)		0.005***
		(0.002)
Social distancing phase (+)		0.006**
		(0.003)
Pre-lockdown mean	0.087	0.087
Change with respect to pre-lockdown levels	-50.6%	-
Year FE	YES	YES
Week FE	YES	YES
Day of the week FE	YES	YES
Census tract FE	YES	YES
Num. obs.	3766905	3766905
R ²	0.097	0.097
Adj. R ²	0.096	0.096

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina) and the Government of the City of Buenos Aires (Argentina) Data Portal.

Notes: The dependent variable is the daily number of reports of thefts, burglaries and larcenies per census tract. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Clustered standard errors by day in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A5: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires at the Census Tract Level (IHS Transformation)

	Model 1	Model 2
Mandatory lockdown (-)	-0.036***	-0.057***
	(0.001)	(0.002)
Authorization of essential workers (+)		0.008***
		(0.002)
Opening of local shops (+)		0.006***
		(0.002)
Opening of shops, permit for runners and recreational outings for		0.012***
children on weekends (+)		(0.002)
Return to phase 1 (strict lockdown) (-)		-0.005**
		(0.002)
Staggered reopening of the city (+)		0.006***
		(0.002)
Opening of bars and outdoor social gatherings (+)		0.004***
		(0.001)
Social distancing phase (+)		0.006**
		(0.002)
Change in % terms	-3.5%	-
Year FE	YES	YES
Week FE	YES	YES
Day of the week FE	YES	YES
Census tract FE	YES	YES
Num. obs.	3766905	3766905
R ²	0.082	0.082
Adj. R ²	0.081	0.081

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina) and the Government of the City of Buenos Aires (Argentina) Data Portal.

Notes: The dependent variable is the IHS transformation of the daily number of reports of thefts, burglaries and larcenies per census tract. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Clustered standard errors by day in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Residential	Commercial	Business	Transportation	Parks	Industrial	Barrios Populares
Average effect of	-39.343***	-73.822***	-15.381***	-16.226***	-8.672***	-0.773***	-4.201***
lockdown	(3.594)	(5.297)	(0.968)	(1.101)	(0.749)	(0.198)	(0.511)
Pre-lockdown mean	91.426	140.224	20.937	24.781	11.323	1.441	6.048
Change with respect to pre- lockdown	-43%	-52.6%	-73.5%	-65.5%	-76.6%	-53.7%	-69.5%
Year DV	YES	YES	YES	YES	YES	YES	YES
Week DV	YES	YES	YES	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES	YES	YES	YES
R^2	0.733	0.772	0.645	0.648	0.472	0.152	0.319
Adj. R ²	0.713	0.756	0.619	0.622	0.433	0.089	0.269
Num. obs.	1035	1035	1035	1035	1035	1035	1035

Table A6: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires for Each Type of Geographical Area

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina), the Government of the City of Buenos Aires (Argentina) Data Portal and the National Ministry of Social Development (Registro Nacional de Barrios Populares). Notes: The dependent variable is the daily number of reports of thefts, burglaries and larcenies. All models consider the period January 2018 through October 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Residential	Commercial	Business	Transportation	Parks	Industrial	Barrios Populares
Average effect of lockdown	-0.758 ^{***} (0.088)	-0.880 ^{***} (0.077)	-1.356 ^{***} (0.061)	-1.128 ^{***} (0.070)	- 1.289 ^{***} (0.105)	-0.489 ^{***} (0.113)	-0.814 ^{***} (0.098)
Change in % terms	-53.1%	-58.5%	-74.2%	-67.6%	-72.4%	-38.6%	-55.7%
Year DV	YES	YES	YES	YES	YES	YES	YES
Week DV	YES	YES	YES	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES	YES	YES	YES
R ²	0.683	0.742	0.729	0.690	0.564	0.146	0.312
Adj. R ²	0.659	0.723	0.709	0.667	0.532	0.083	0.261
Num. obs.	1035	1035	1035	1035	1035	1035	1035

Table A7: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires for Each Type of Geographical Area (IHS Transformation)

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina), the Government of the City of Buenos Aires (Argentina) Data Portal and the National Ministry of Social Development (Registro Nacional de Barrios Populares). Notes: The dependent variable is the IHS transformation of the daily number of reports of thefts, burglaries and larcenies. All models consider the period January 2018 through October 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Outside	of homes	Inside o	f homes
	Model 1	Model 2	Model 3	Model 4
Mandatory lockdown (-)	-153.378***	-233.888***	0.348	-1.343
	(9.298)	(10.622)	(1.035)	(1.048)
Authorization of essential workers (+)		34.066***		1.386
		(9.450)		(0.847)
Opening of local shops (+)		27.352***		-0.651
		(8.749)		(0.881)
Opening of shops, permit for runners and		44.455***		2.302***
recreational outings for children on weekends (+)		(7.556)		(0.852)
Return to phase 1 (strict lockdown) (-)		-21.259**		0.313
		(9.255)		(1.098)
Staggered reopening of the city (+)		18.440**		-1.627
		(7.836)		(1.060)
Opening of bars and outdoors social gatherings (+)		20.140**		-0.183
		(8.847)		(0.677)
Social distancing phase (+)		26.655**		-2.301
		(12.476)		(1.803)
Pre-lockdown mean	301.387	301.387	6.287	6.287
Change with respect to pre-lockdown levels	-50.9%	-	5.5%	-
Year DV	YES	YES	YES	YES
Week DV	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES
R ²	0.812	0.555	0.843	0.562
Adj. R ²	0.797	0.522	0.831	0.527
Num. obs.	1035	1035	1065	1065

Table A8: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires Committed Outside and Inside of Homes

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the daily number of reports of thefts, burglaries and larcenies. Models 1 and 2 include all thefts and larcenies not committed in a household, and models 3 and 4 include those committed in a household. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 and 2 consider the period January 2018 through October 2020. Models 3 and 4 use information until November 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A9: Estimation of the Detectar Program on Thefts, Burglaries and
Larcenies Reports in <i>Barrios Populares</i> in the City of Buenos Aires

	Model 1	Model 2
Lockdown	-0.482***	-0.159***
	(0.097)	(0.031)
Detectar Program	-1.492***	-0.172***
	(0.244)	(0.066)
Lockdown coefficient in % terms	-	-14.7%
Detectar coefficient in % terms	-	-15.8%
Barrio Popular FE	YES	YES
Year FE	YES	YES
Week FE	YES	YES
R^2	0.705	0.747
Adj. R ²	0.700	0.742
Num. obs.	6336	6336

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina) and the National Ministry of Social Development (Registro Nacional de Barrios Populares). Notes: The dependent variable of Model 1 is the number of reports of thefts, burglaries and larcenies per week per *barrio popular*. The dependent variable of Model 2 is the IHS transformation of the number of thefts and larceny reports per week per *barrio popular*. Both models consider the period January 2018 through September 2020. HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A10: Estimation of the Effect of the Detectar Program on Thefts, Burglaries and
Larcenies in Barrios Populares in the City of Buenos Aires Relative to Their
Surrounding Neighborhoods

	All Barrios Populares	Main Barrios Populares
Lockdown	0.002	0.015 [*]
	(0.002)	(0.009)
Detectar Program	-0.001	-0.018 [*]
	(0.002)	(0.009)
Barrio Popular FE	YES	YES
Year FE	YES	YES
Week FE	YES	YES
R ²	0.569	0.500
Adj. R ²	0.561	0.455
Num. obs.	6316	864

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina) and the National Ministry of Social Development (Registro Nacional de Barrios Populares).

Notes: The dependent variable is the weekly ratio between the number of thefts, burglary and larceny reports per *barrio popular* relative to the number of such reports in the neighborhood where the *barrio popular* is located. Model 1 includes all *barrios populares*, whereas model 2 only includes the main *barrios populares* where the *Detectar* Program was introduced separately from the rest of the overall commune (*Barrio 31, Barrio 21-24, Barrio 1-11-14, Barrio 20, Barrio 15* and *Ramón Carrillo*). Both models consider the period January 2018 through September 2020. HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Day		Night	
	Model 1	Model 2	Model 3	Model 4
Mandatory lockdown (-)	-117.919***	-186.340***	-45.296***	-58.429***
	(9.471)	(12.239)	(3.191)	(4.398)
Authorization of essential workers (+)		28.723**		6.307
		(12.074)		(5.493)
Opening of local shops (+)		20.373**		6.330
		(10.293)		(6.338)
Opening of shops, permit for runners		40.234***		7.596
and recreational outings for children on weekends (+)		(9.895)		(6.373)
Return to phase 1 (strict lockdown) (-)		-11.138		-12.690
		(12.244)		(11.021)
Staggered reopening of the city (+)		8.340		9.742
		(10.299)		(10.946)
Opening of bars and outdoors social		22.121**		-2.153
gatherings (+)		(9.611)		(6.416)
Social distancing phase (+)		63.178**		-40.215*
		(27.085)		(23.991)
Pre-lockdown mean	230.546	230.546	73.939	73.939
Change with respect to pre-lockdown levels	-51.1%	-	-61.3%	-
Year DV	YES	YES	YES	YES
Week DV	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES
R ²	0.705	0.733	0.554	0.552
Adj. R ²	0.683	0.711	0.520	0.516
Num. obs.	1035	1065	1035	1065

Table A11: Estimation of the Effect of the Lockdown on the Number of Daily Reports for Thefts, Burglaries and Larcenies in the City of Buenos Aires for Day and Nighttime

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the daily number of reports of thefts, burglaries and larcenies during the day or night (10pm - 6am). The first and second models include crimes committed during the day and the third and fourth include crimes committed at night. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Models 1 and 3 consider the period January 2018 through October 2020. Models 2 and 4 use information until November 2020. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A12: Estimation of the Effect of the Lockdown on the Number of Weekly Report	S
of Homicides in the City of Buenos Aires (IHS transformation)	

	Model 1	Model 2
Mandatory lockdown (-)	0.100	-0.155
	(0.272)	(0.463)
Authorization of essential workers (+)		0.384
		(0.436)
Opening of local shops (+)		-0.087
		(0.548)
Opening of shops, permit for runners and recreational outings for		-0.438
children on weekends (+)		(0.501)
Return to phase 1 (strict lockdown) (-)		1.204**
		(0.502)
Staggered reopening of the city (+)		-0.202
		(0.319)
Opening of bars and outdoor social gatherings (+)		-0.839**
		(0.333)
Year DV	YES	YES
Month DV	YES	YES
R ²	0.117	0.167
Adj. R ²	0.029	0.044
Num. obs.	156	156

Notes: The dependent variable is the IHS transformation of the weekly number of homicide reports. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Model 1 considers the period January 2018 through October 2020. Model 2 uses information until November 2020. Models include year- and month-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	Homicides outside of barrios populares		cides outside of Homicides rios populares barrios po	
	Model 1	Model 2	Model 3	Model 4
Mandatory lockdown (-)	0.90*	0.32	-0.59	-0.86*
	(0.47)	(0.76)	(0.45)	(0.45)
Authorization of essential workers (+)		0.48		0.52**
		(0.72)		(0.23)
Opening of local shops (+)		-0.35		0.43
		(0.69)		(0.66)
Opening of shops, permit for runners and		-0.57		-0.56
recreational outings for children on weekends (+)		(0.76)		(0.63)
Return to phase 1 (strict lockdown) (-)		2.12 [*]		0.55
		(1.22)		(0.54)
Staggered reopening of the city (+)		-0.18		-0.42
		(1.23)		(0.52)
Opening of bars and outdoor social gatherings (+)		-0.83		-0.94***
		(0.99)		(0.30)
Social distancing phase (+)		-0.38		0.89
		(0.64)		(0.64)
Pre-lockdown mean	1.65	1.65	0.54	0.54
Change with respect to pre-lockdown mean	54.7%		-110.17%	
Year DV	YES	YES	YES	YES
Month DV	YES	YES	YES	YES
R ²	0.11	0.14	0.11	0.18
Adj. R ²	0.02	0.01	0.01	0.04
Num. obs.	152	152	152	152

Table A13: Estimation of the Effect of the Lockdown on the Weekly Number of Homicides in the City of Buenos Aires for Crimes Committed Outside or Inside *Barrios Populares*

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina) and the National Ministry of Social Development (Registro Nacional de Barrios Populares).

Notes: The dependent variable is the weekly number of homicide reports. A (+) refers to a relaxation of the isolation measures, while a (-) reflects more restrictions. Models 1 and 3 consider the period January 2018 through October 2020. Models 2 and 4 use information until November 2020. Models include year- and month-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

Table A14: Est	mation of the Effect of the Lockdown on the Number of Daily Detainees					
in the City of Buenos Aires by Address of the Detainee (IHS Transformation)						
-						

-

	City of Buenos Aires	Province of Buenos Aires	Homeless	Different commune	Same commune
Mandatory	-0.326***	-0.757***	-0.191**	-0.385***	-0.182**
lockdown	(0.040)	(0.080)	(0.081)	(0.047)	(0.087)
Change in % terms	-27.8%	-53.1%	-17.3%	-32%	-16.6%
Year DV	YES	YES	YES	YES	YES
Week DV	YES	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES	YES
R ²	0.368	0.652	0.491	0.489	0.443
Adj. R ²	0.294	0.612	0.431	0.429	0.378
Num. obs.	670	670	670	670	670

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The dependent variable is the IHS transformation of the number of detainees each day. The first model only includes detainees with an address in the City of Buenos Aires; the second model contains only detainees with an address in the Province of Buenos Aires; the third model only computes homeless detainees; the fourth and fifth models consider detainees with an address in a different commune or the same commune of the City of Buenos Aires, respectively. All models consider the January 2019 through October 2020 period. Models include year-, week- and day of the week-specific dummy variables (DV). HAC standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.

	All crimes	Residence of Detainee	Type of Crime								
		City of Buenos Aires	Drug trafficking	Theft	Motorbike thieves	Larceny	Drug dealing	Sexual assault	Homicide	Resistance to authorities	Others
Mandatory lockdown	-16.874***	-1.350	3.292	-16.558***	-17.887	-43.776***	64.92***	-71.279***	-36.678	-53.113***	-23.748***
	(2.694)	(0.902)	(5.297)	(5.921)	(22.228)	(9.264)	(21.635)	(23.708)	(37.925)	(13.567)	(4.740)
Pre-lockdown mean	108.714	37.595	102.748	119.552	88.566	122.397	158.345	98.662	119.473	89.702	102.298
Change with respect	-15.52	-3.59	3.2 percent	-13.85	-20.2	-35.77	41	-72.25	-30.7	-59.21	-23.21
to pre-lockdown mean	percent	percent		percent	percent	percent	percent	percent	percent	percent	percent
Year DV	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Week DV	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Day of the week DV	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
R ²	0.013	0.068	0.026	0.023	0.228	0.034	0.308	0.208	0.277	0.058	0.023
Adj. R ²	0.011	0.065	0.019	0.016	0.065	0.014	0.252	0.063	-0.033	0.024	0.018
Num. obs.	38524	25787	9638	8919	402	3379	931	454	234	1993	12574

Table A15: Estimation of the Effect of the Lockdown on the Distance Travelled by the Detainee from its Home Address to Commit the Crime

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina). Notes: The dependent variable is the distance (in blocks) between the center of the administrative unit (commune -for detainees residing the City of Buenos Aires- or *Partido* -for those in Buenos Aires province) of the detainee's home address and the exact location of the crime. All models consider the period January 2019 through October 2020. Models include year-, week- and day of the week-specific dummy variables (DV). Robust standard errors in parentheses. *** p < 0.01; ** p < 0.05; * p < 0.1.



Figure A1: Daily Vehicles Counted by Government Traffic Cameras

Source: Authors' calculations based on data from the Government of the City of Buenos Aires (Argentina) Data Portal. Notes: The solid black line is the 7-day moving average (backward-looking) of the daily number of vehicles counted by the government's traffic cameras. The dashed line indicates the beginning of the mandatory isolation period.

Figure A2: Daily Number of One-way Tickets Sold in the Public Transport System



Source: Authors' calculations based on data from the Government of the City of Buenos Aires (Argentina) Data Portal.

Notes: The solid black line is the 7-day moving average (backward-looking) of the daily number of one-way tickets sold in the public transport system. The dashed line indicates the beginning of the mandatory isolation period.



Figure A3. Mobility in the City of Buenos Aires: Google COVID-19 Community Mobility Report

Source: Authors' elaboration based on data from the Google COVID-19 Community Mobility Reports. Changes in mobility are calculated using the same kind of aggregated and anonymized cellular use data utilized to show popular times for places in Google Maps. Information obtained from users who have opted-in to Location History for their Google Account.

Notes: Lines represent the 7-day moving average (backward-looking) of the percent change in mobility with respect to the baseline period. The baseline is the median value, for the corresponding day of the week, during the 5-week period (Jan 3–Feb 6, 2020). The dashed line indicates the beginning of the mandatory lockdown.



Figure A4. Number of Covid-19 Cases per 100,000 People

Source: Authors elaboration based on data from the Dirección Nacional de Epidemiología y Análisis de Situación de Salud.

⁷⁻day moving average — Argentina ---- CABA



Figure A5. Number of Covid-19 Related Deaths per 100,000 People

7-day moving average — Argentina ---- CABA

Source: Authors' elaboration based on data from the Dirección Nacional de Epidemiología y Análisis de Situación de Salud.



Figure A6. Classification of the City of Buenos Aires According to Land Use

Sources: The government of the City of Buenos Aires publishes the city blocks' categorization according to the urban planning code: https://www.buenosaires.gob.ar/planificacion/informacion-para-tu-proyecto/dataset (accessed October 2020): 2017 30. and the land use survev: https://data.buenosaires.gob.ar/dataset/relevamiento-usos-suelo (accessed October 30, 2020); Data from Properati is available in: https://www.properati.com.ar/data (accessed October 18, 2020). We used information for the second semester of 2019.

Notes: To systematically classify the location where each crime was committed into seven land use categories (residential, commercial, business/offices, transportation, parks, industrial, and *barrios populares*), we trained a machine-learning (XGBoost) model with points in the City classified using the urban planning code, the 2017 land use survey, and data from properties listed for sale and rent on Properati, a major real estate marketplace. Over 300 points in the city were manually classified for training and validating the model. The model cross-validated accuracy, a measure of performance, exceeded 70 percent. We used the model to classify 25,000 random points in the City of Buenos Aires and then assign each reported crime according to its nearest neighbor's classification.



Figure A7. Daily Reported Property Crimes in the City of Buenos Aires for Each Commune

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Notes: The values (y-axis) are the natural logarithm of the 15-day moving average of the number property crimes in each commune of the City of Buenos Aires..



Figure A8. Daily Reported Property Crimes in the City of Buenos Aires for Each Type of Area

Source: Authors' calculations based on data from the Ministry of Justice and Security of the City of Buenos Aires (Argentina).

Note: The values (y-axis) are the natural logarithm of the 15-day moving average of the number property crimes in each type of area of the City of Buenos Aires.